Author’s response to reviews

Title: Listening to health workers: lessons from Eastern Uganda for strengthening the programme for the prevention of mother-to-child transmission of HIV

Authors:

Joseph Rujumba (jrujumba@yahoo.com)
James K. Tumwine (kabaleimc@gmail.com)
Thorkild Tylleskår (thorkild.tylleskar@cih.uib.no)
Stella Neema (sheisim@yahoo.com)
Harald K. Heggenhougen (kheggenh@bu.edu)

Version: 4 Date: 4 January 2012

Author’s response to reviews: see over
Date: 4th January, 2012

The Editor-in-Chief  
BMC Health Services Research  

Dear Editor,

**Response to editors – Formatting changes on our manuscript titled:** Listening to health workers: lessons from Eastern Uganda for strengthening the programme for the prevention of mother-to-child transmission of HIV

Thank you for the acceptance notice and formatting changes required in regard to our manuscript. We have effected the formatting changes as follows:

1) Cited ref 34 in the text see page 28, 1st paragraph, 4th sentence.
2) Reference 26 has also been corrected; it now starts as UBOS …
3) Checked all references and confirm that they are cited in numerical order in as cited in text.
4) We have proof read the manuscript.

Once again, thank you for the opportunity to publish with you.

Yours,

Joseph Rujumba  
Corresponding author.